Author's response to reviews

Title: Comparison of two protective lung ventilatory regimes on oxygenation during one-lung ventilation: A randomized controlled trial

Authors:

Felix R. Montes (felixmontes@etb.net.co)
Daniel F. Pardo (pipepardo@me.com)
Hernan Charris (hcharris@hotmail.com)
Luis J. Tellez (ltellez@carioinfantil.org)
Juan C. Garzón (jcgarzon@mac.com)
Camilo Osorio (cirugiatroracica@cardioinfantil.org)

Version: 3  Date: 1 November 2010

Author's response to reviews: see over
November 1, 2010

The Independent Editorial Production Team
The Journal of Cardiothoracic Surgery

Re: Manuscript: 1334497874436849 Comparison of two protective lung ventilatory regimes on oxygenation during one-lung ventilation. A randomized controlled trial.

Dear Sirs:

Thank you for your response and the acceptance for publication of the article. The manuscript has been revised making all of the changes suggested to the format.

Per your request, I am including a point-by-point description of the changes made.

Major revisions:

1. Affiliations: All affiliations were written in full according to the suggested format.
2. Footnotes: Following your recommendation we incorporate the footnote on page 6 into the text of the manuscript.

Minor revisions:

1. We removed the highlighting from the manuscript.
2. The heading “Competing interest” was change to “Competing interests” (page 11).
3. The blank page on page 18 was removed.

Thank you very much for your time and attention.

Sincerely,

Felix R. Montes, M.D.
Anesthesiology Department
Fundación Cardio Infantil
Universidad del Rosario
Bogota, Colombia